Author's response to reviews

Title: Revision in total hip arthroplasty using a ceramic revision head: Two year results of eight patients

Authors:

Dirk Ganzer (ganzerdr@t-online.de)
Lutz Forke (forke@ms-arn.de)
Ulrich Irlenbusch (irlenbusch@ms-arn.de)

Version: 2
Date: 18 June 2014

Author's response to reviews: see over
Dear Dr. Kidd,

I am delighted to resubmit our paper: “Revision total hip arthroplasty using a ceramic revision head with a retained well-fixed femoral component: Two year results in eight patients” for consideration for publication in the *Journal of Medical Case Reports*.

The manuscript has been revised according to the reviewers’ feedback, and we feel that all of their points have been appropriately addressed. This manuscript has not been previously published and is not under consideration in the same or substantially similar form in any other peer-reviewed media. All authors have read and approved the manuscript. Each author believes that the manuscript represents honest work.

Editorial support was provided by Fishawack Archimed AG, Zofingen, Switzerland, and was paid for by Mathys Ltd., Bettlach, Switzerland.

We sincerely believe that the data we are presenting will be of great interest to readers of the *Journal of Medical Case Reports*, and hope that you agree this paper is now suitable for publication.

Yours sincerely

Dr. med. Dirk Ganzer (on behalf of all authors)

ganzerdr@t-online.de